Author’s response to reviews

Title: Wound Infections Following Implant removal below the knee: the effect of antibiotic prophylaxis; the WIFI-trial, a Multi-Centre Randomized Controlled Trial

Authors:

M Backes (m.backes@amc.nl)
S A Dingemans (s.a.dingemans@amc.uva.nl)
N WL Schep (n.w.schep@amc.uva.nl)
F W Bloemers (fw.bloemers@vumc.nl)
B van Dijkman (bydijkman@flevoziekenhuis.nl)
F P Garssen (frgar@zha.nl)
R Haverlag (r.haverlag@olvg.nl)
P Heres (pheres@wlz.nl)
H M Hoogendoorn (j.hoogendoorn@mchaaglanden.nl)
V Postma (v.postma@mcgroep.nl)
E Ritchie (e.ritchie@rijnland.nl)
W H Roerdink (roerdink@dz.nl)
J B Sintenie (sintenie@hotmail.com)
N MR Soesman (nsoesman@vlietlandziekenhuis.nl)
N L Sosef (nsosef@spaarnziekenhuis.nl)
E Ph Steller (e.steller@slaz.nl)
B A Twigt (b.twigt@bovenij.nl)
A H van der Veen (alexander.vd.veen@catharinaziekenhuis.nl)
R van Velde (rvanvelde@tergooi.nl)
D I Vos (dvos@amphia.nl)
M R de Vries (m.devries@rdgg.nl)
J Winkelhagen (j.winkelhagen@westfriesgasthuis.nl)
J C Goslings (j.c.goslings@amc.nl)
T Schepers (t.schepers@amc.nl)

Version: 2
Date: 6 January 2015

Author’s response to reviews: see over
Dear Eden Nonato,

I hereby would like to reply on your additional requests considering the manuscript entitled ‘Wound Infections Following Implant removal below the knee: the effect of antibiotic prophylaxis; the WIFI-trial, a Multi-Centre Randomized Controlled Trial’ (MS: 1330237930138234).

Yours sincerely,

Manouk Backes
Corresponding author

Additional request formatting:

1. Please include your abstract in the main manuscript file, and please also add a Discussion in your abstract. For more information, please click on the following link: http://www.biomedcentral.com/bmcsurg/authors/instructions/studyprotocol#formatting-abstract.

   We have added the abstract to the main manuscript file, and have added a discussion to the abstract.

2. Please add a discussion section to elucidate the importance of this clinical trial. For more information, please click on the following link: http://www.biomedcentral.com/bmcsurg/authors/instructions/studyprotocol#formatting-discussion.

   We have added a discussion to the main manuscript.

3. Kindly include the email of each author in the title page. For more information, please click on the following link: http://www.biomedcentral.com/bmcsurg/authors/instructions/studyprotocol#formatting-title.

   We have added the email addresses of all the authors.

4. Please add an Acknowledge section following the Authors’ Contribution section. For more information, please click on the following link: http://www.biomedcentral.com/bmcsurg/authors/instructions/studyprotocol#formatting-acknowledgements.

   We have added an Acknowledge section.

5. Please kindly remove the Ethics approval and Funding document from your additional files, as we do not require that these be published alongside your manuscript.

   We have removed the Ethics Approval and Funding document from the additional files. These were added as the following is stated on your website: ‘Articles concerning proposed research will usually be considered for publication without peer review if the study has received ethics approval and a grant from a major funding body. Proof of both ethics and funding will be required and we recommend that authors provide the relevant documentation on submission.’